Author’s response to reviews

Title: Measuring liver fat fraction with complex-based chemical shift MRI: the effect of simplified sampling protocols on accuracy

Authors:

Alexander Procter (ajprocterac@gmail.com)

Julia Sun (julia.yusun@nnuh.nhs.uk)

Paul Malcolm (Paul.malcolm@nnuh.nhs.uk)

Andoni Toms (andoni.toms@nnuh.nhs.uk)

Version: 2 Date: 09 Jan 2019

Author’s response to reviews:

RE Editors comments:

1. As per reviewer 1’s suggestions I have converted inter-observer agreement to Spearman's rank correlation coefficient. All mentions of ICC have been changed. As reviewer 1 expected, this statistical change has not affected the conclusions.

2. Done

3. Consent statement also added to Declarations section

4. Done

5. Done

6. Done

Thank you

Alex Procter